Author's response to reviews

Title: Role of SMC1A overexpression as a predictor of poor prognosis in colorectal cancer

Authors:

Jianwei Wang (jianweidr@163.com)
Shaojun Yu (shaojundr@163.com)
Liming Cui (seniorever@163.com)
Wenhui Wang (wwh1023@126.com)
Jun Li (junlidr@163.com)
Ke Wang (kewangdr@163.com)
Xinyuan Lao (laoxydr@163.com)

Version: 2 Date: 18 November 2014

Author's response to reviews: see over
Dear Editor,

Please find a revised manuscript entitled “Role of SMC1A overexpression as a predictor of poor prognosis in colorectal cancer”. Accordingly, this manuscript has been revised in response to every concern or comment raised by the editors. We hope the revised manuscript will meet the journal’s standard. Below you will find our point-to-point responses.

1). Please provide Line numbering.

Line numbers have been provided in the revised manuscript.

2). Please format Title page.

Title page has been formatted at the front of our manuscript.

3). Please provide Authors’ Contributions section.

Authors’ Contributions section has been provided in the revised manuscript.

We thank you for your time and consideration of this manuscript for publication in BMC Cancer.

Yours sincerely,

Dr. Xinyuan Lao

Holly Lab Shanghai, 333 Guiping Road, Xuhui District, Shanghai 200233, China

Tel.: +8602134692658

Fax: +8602134692657

E-mail: laoxydr@163.com